Impact of Corporate Governance on Related Party Transactions in Family-Owned Firms in Pakistan

Fazli Azima, Mohd Zulkhairi Mustaphab, Fauzi Zainirc

Abstract: This is an empirical study to examine the impact of corporate governance (CG) variables, namely independent directors index (IDI), family directorship (FD), and family ownership (FO), on the types of related party transactions (RPTs) that prevail in family-owned firms in developing countries. The focus of this study is on Pakistan whereby it was found major shareholders of Pakistani family-owned firms expropriate resources through RPTs. This study analysed panel data of 108 firms of family-owned firms listed on the Karachi stock exchange from 2004 to 2014 after the introduction of CG codes in 2002. The study contributes to literature by categorising all RPTs into three types - RPTb, RPTe, and RPTo. It also develops an index of independence directors comprising three dimensions, namely board composition, financial expertise, and tenure of the independent non-executive director. Different panel least squares models for different RPTs have been employed to examine the relationship between RPTs and CG variables. This study found that 90% of familyowned firms in Pakistan scored low for independent director's index. Hence, CG is weaker in Pakistani family-owned firms where major shareholders expropriate resources through RPTb and RPTe. Further, CG variables, namely IDI and FD are negatively related to RPTs and the concentration of family ownership of firms has a negative relationship with RPTs, and this has a negative tendency for the resource expropriation of family-owned firms. The study concludes the negative relationship of corporate governance and concentration of major shareholder exploit the interest of minority shareholders in Pakistani family-owned firms.

Keywords: Corporate Governance Related party transactions, Pakistani family-owned business, major shareholder, index of independent directors. *JEL classification:* G34, G38, G180

Article Received: 05 April 2017; Article Accepted: 08 March 2018

1. Introduction

Studies have documented how Abusive Related Party Transactions (RPTs) are methods used by insider¹ shareholders to exploit outsider shareholders

^a Department of Accountancy, Faculty of Business and Accountancy, University of Malaya. *Email:* Fazli.azim@siswa.um.edu.my

^b Corresponding author. Department of Accountancy, Faculty of Business and Accountancy, University of Malaya. *Email: zulkhairi@um.edu.my*

^c Department of Accountancy, Faculty of Business and Accountancy, University of Malaya. *Email:* zfauzi@um.edu.my

(Ryngaert & Thomas, 2012). Investigations have shown that abusive RPTs are responsible for financial scandals, such as Enron, WorldCom, Parmalat, Adelphia Communications, Coloroll, Maxwell Group, Nortel, Polly Peck, Royal Ahold, and Satyam, (Zalewska, 2014). This fraudulent activity through RPTs is of great concern for regulators and investors as they have advantages such as saving transaction costs and improving operating efficiency of the companies (Ge, Drury, Fortin, Liu, & Tsang, 2010). Hence, corporate governance (CG) is concerned with protecting interest of owners and implemented in an organisation (Sarbah & Xiao, 2015). Almost all countries have developed their own set of codes for CG, which could also function as guidelines. The CG codes were introduced in late 20th century and received good response from companies (Cadbury, 1992; Remuneration & Greenbury, 1995; Sarbanes, 2002).

Conflict of interest between major and minority shareholders is common in East Asia and in the West where major shareholders control the firms (Claessens, Djankov, Fan, & Lang, 2002). The controlling shareholder have the capability to expropriate the interest of minority shareholders (Claessens, Djankov, & Lang, 2000; La Porta, Lopez-de-Silanes, Shleifer, & Vishny, 1999) and "tunneling" or expropriation of resources" may be an obvious source of expropriation (Johnson, La Porta, de Silanes, & Shleifer, 2000). Therefore, major shareholders have the opportunity to expropriate funds from bottom to up through the pyramidal structure due to differences between cash flow and control rights, and this will make them wealthier at the expense of minority shareholders, (Riyanto & Toolsema, 2008). This tunnelling process therefore will negatively impact on interest of minority shareholders, who generally gain less from their shareholding. Such resource transfer may be costly for minority shareholders, as well as decreases transparency of the whole economy, shows biased accounting figures, and makes the examination of a company's true performance difficult. Bertrand, Mehta, and Mullainathan (2002) found a significant amount of tunnelling in India in transfer pricing contracts and asset sales or even outright cash appropriation. The same situation of exploitation of resources occur in family-owned firms in Pakistan where major shareholder transfer resources at the expense of minority shareholder ((Bhutta, Knif, & Sheikh, 2016; Ullah & Shah, 2015).

This conflict of interest is consistent with the model proposed by Berle and Means (1932) which showed the relationship between ownership structure and performance of firms with diverse ownership (where every owner holds a small percentage of total ownership). This n is supported by Agency theory of Jensen and Meckling (1976), where corporate governance separates company ownership from management. Corporate governance emerged because of two issues: (i) agency issue and (ii) trade cost. Agency issues arise when the interests of owners and management are in conflict. Thus, owner (principal) and management (agent) look for common interests. Trade cost arises when the agreement between owners and management fail to consider all future uncertain events. This situation could likely lead to opportunistic behaviour by management (Shleifer & Vishny, 1997).

The concentration of family ownership is a major issue in Pakistani family-owned firms (Ali, Tahir, & Nazir, 2015; Hussain & Shah, 2015). The percentage of concentrated family ownership is almost half of the corporate ownership held by large or concentrated owners (Javid & Iqbal, 2008). This concentration of family ownership has high negative impact on company performance and reduces corporate efficiency and economic development (Afgan, Gugler, & Kunst, 2016). This situation may result in the expropriation of resources and exploitation of minority shareholder interests by large shareholders (Abbas, Naqvi, & Mirza, 2013). Corporate governance has no effect on dividend pay-out and firm value, although dividend pay-out and firm value are significantly related (Ali et al., 2015). The controlling shareholders in family-owned firms expropriate funds from lower to upper level firms through pyramidal structure. The interests of minority shareholders are exploited due to this fund expropriation. This resource transfer supports the Model of Berle and Means (1932) Agency theory of Jensen and Meckling (1976) and Conflict of interest views of Gordon, Henry, and Palia (2004). Despite the existence of CG codes, the performance of family owned-firms decreases (Afza & Nazir, 2015). The role of controlling shareholders in Pakistan varies due to the preference of firm owners (Tahir & Sabir, 2015). The International Finance Corporation (IFC,2007)² highlighted weaknesses of CG governance in Pakistan. IFC (2007) emphasised that the corporate board has low percentage of experienced personnel and low or no protection for minority shareholders. Lack of law enforcement with respect to investor right is rampant, as courts are swamped with cases and prosecution of such cases is costly and timeconsuming in deciding on the settlement. While listed companies provide adequate and timely disclosure, several groups in the manufacturing sector and state-owned firms do not follow rules and regulations. As penalty is low for not providing full disclosure, the company is not motivated to follow rules and regulations. The IFS (2007) further cited the disclosure issue of conflict of interests and related party transactions. Similarly, few family-owned firms influence and control resources. These family-owned firms are usually involved in the expropriation of resources at the expense of minority shareholders, (Ibrahim, 2006). A developing country like Pakistan is an example of a complex situation in relation to examining these issues because capital markets are under developed with low stock market capitalisation and foreign direct investment, (Gohar & Karacaer, 2009). Hence, speculation and corruption activities are heightened.

Empirical evidence shows that affiliated firms have poorer performance than unaffiliated firms. The average values of the Tobin's Q, ROA, and OPROA of affiliated firms are also significantly lower. These values suggest monitoring group activities of family-owned firms through outsiders. This factor reduces agency problems and diminishes performances of family-owned firms than unaffiliated firms. Faccio, Lang, and Young (2001) argued the existence of agency problems in Asian firms employing CG and engaging in a political environment. A study on the Pakistan expropriation of resources in was conducted before implementation of corporate governance codes in 2002, (SECP codes, 2002). Ikram and Naqvi (2005) showed the expropriation of assets of 86 firms belonging to family-owned firms for 10 years (1993-2003). The authors concluded the existence of tunnelling in family-owned firms. Other issues were governing board, independence of board, no balance of power in the board, non-executive directors in firm succession, trust, and confidence of the investors, and disclosure of family-owned firms (Ameer, 2013). All of these may create problem for minority shareholders and other stake holders (Mehboob, Tahir, & Hussain, 2015).

The three objectives of this study are as follows. First, it investigates the effect of independent director index on RPTs. Second, it determines the effect of family directorship on RPTs. Finally, it examines the effect of family ownership on RPTs.

The contributions of this study are first, it contributes to literature by suggesting ways to minimise the exploitation of the minority shareholder interests by major shareholders in Pakistani family-owned firms through their high ownership concentration, i.e., agency theory (type II), conflict of interest between major and minority shareholders (Jensen & Meckling, 1976), and conflict of interest views (Gordon et al., 2004). Second, it contributes to corporate governance literature in the context of developing countries, such as Pakistan, by empirically examining the impact of independent non-executive directors in preventing abusive related party transactions. Third, it develops an index of independent directors (IDI) that examines the effect of firm performance in family-owned firms. Most studies have reported on the importance of the independent director in terms of composition and financial expertise in family-owned firms. However, this study added one more dimension to the independent director variable (i.e., tenure). The final index consists of three attributes of independent directors, namely composition, financial expertise, and tenure. The independent director plays a key role in mitigating the resource transfer by major shareholders in family-owned firms. The role of independent directors (IDI) includes critical issues. The independency of independent director must be examined based on the above-mentioned dimensions because most family-owned firms in Pakistan fall in the lowest level of IDI (Figure 1 in Appendix C). An independent director is mainly responsible for mitigating abusive RPTs. Fourth, it contributes empirically by testing the impacts of IDI, FD and FO on types of RPTs, namely, RPTb, RPTe, and RPTo, prevailing in Pakistani family-owned firms where major shareholders expropriate resources through abusive RPTs (Agrawal & Knoeber, 2012; Azeez, 2015; Baysinger & Butler, 1985; Kang & Shivdasani, 1995). Fifth, it contributes to literature by identifying new categorisations of RPTs. Gordon et al. (2004), Cheung, Rau, and Stouraitis (2006), Lei and Song (2008), Cheung, Qi, Rau, and Stouraitis (2009), Kohlbeck and Mayhew (2010), Jian and Wong (2010), Ryngaert and Thomas (2012), Srinivasan (2013) and Williams and Taylor (2014) have identified various RPTs in firm groups. However, this research identified all RPTs between controlling shareholder companies and subsidiaries and categorised these RPTs in 12 different types of RPTs in Pakistani familyowned firms, which were further sub-categorised. The details of this categorisation are shown in Appendix "A." In addition, the two types of RPTs, namely, benefit- and expense-based transactions were identified. The remaining 10 types of RPTs have been discussed by various researchers along with their implications and categorised as other types of RPTs.

The rest of the study is organised as follows. Section 2 provides literature review and hypotheses. Section 3 describes the methodology while Section 4 discusses the empirical results. Section 5 provides policy implications and conclusion.

2. Literature review and hypotheses 2.1 Literature review

According to US GAAP, RPTs is defined as "transactions between a company and its subsidiaries, affiliates, principal owners, officers or their families, directors or their families, or entities owned or controlled by its officers or their families"³ According to the International Accounting Standards (IAS), RPTs "can be a person, an entity, or an unincorporated business"⁴ This definition has two categories. The first recognises " a person, or a close member of that person's family, being a related party from the perspective of the reporting entity." The second is "an entity related to the reporting entity." Similarly, Gordon et al. (2004) explored two alternative perspective of RPTs. The first is conflict of interest transactions and the second is efficient transactions. First, conflict of interest transactions can be termed as abusive, which is consistent with agency theory of Jensen and Meckling (1976) and Model of Berle and Means (1932). This conflict is potentially harmful to shareholder interests (Aharony, Wang, & Yuan, 2010; Cheung et al., 2006; Gordon et al., 2004; Jiang, Lee, & Yue, 2010). Second, efficient transaction extends the

transaction cost concept developed by Coase (1937) and Williamson (1975) and shows that related party transactions will benefit, not harm shareholders (Chang & Hong, 2000; Jian & Wong, 2010; Khanna & Palepu, 2000; Stein, 1997).

Corporate governance practices and structures have witnessed enormous changes over the last two decades. Most firms in developing and developed countries have concentrated ownerships (La Porta, Lopez-de-Silanes, Shleifer, & Vishny, 2000). These controlling shareholders normally use their stakes of ownership concentration, exercise control rights that surpass their cash flow rise, and provide more opportunities to insiders to expropriate outside shareholders through various firm operations and financing decisions (Bertrand et al., 2002; Claessens et al., 2002; Faccio et al., 2001; Gopalan & Jayaraman, 2012; Johnson et al., 2000; La Porta, Lopez- de- Silanes, Shleifer, & Vishny, 2000; Porta, Lopez- de- Silanes, Shleifer, & Vishny, 2002). This exploits the wealth of minority shareholders through tunnelling, (Bae, Kang, & Kim, 2002; Buysschaert, Deloof, & Jegers, 2004).

Corporate governance becomes more significant in family-owned firms, specifically in developing countries where most firms are dominated by families (Claessens et al., 2000; La Porta, Lopez-de-Silanes, et al., 2000). The ownership structure of these family-owned firms may have cross shareholding or pyramidal structures in which members of the board of directors belong to the same family (Javid & Iqbal, 2010a). The controlling family is the major owner and controller, whereas the immediate and distant family-members help operate various firms within family-owned firms (Ghani, Haroon, & Ashraf, 2010). When family-owned firms grow, conflict of interest arises among the owners, managers, and employees (Bennedsen & Wolfenzon, 1998; Porta, Lopez- de- Silanes, & Shleifer, 1999). A good corporate governance system brings right policies to manage such conflict of interest (Sarbah & Xiao, 2015). Gohar & Karacaer, 2009 have also highlighted these issue in Pakistani capital markets because they are under developed, as indicated by their low level of stock market capitalisation and foreign direct investment. They further noted that there is lack of law enforcement and high speculation activities and corruption in country. The few family-owned firms are powerful and dominate the economic landscape. Controlling shareholders in Pakistani family-owned firms expropriate funds from bottom- to upper-type firms through pyramidal ownership. The expropriation of resources occurs due to high percentage of concentrated ownership that almost half of the corporate ownership is held by large or concentrated owners (Javid & Iqbal, 2008). This high ownership concentration has negatively impacts on company performance, namely expropriation of resources and exploitation of minority shareholders by large shareholders (Abbas et al., 2013).

Therefore, corporate governance codes are developed to safeguard the right of shareholders in an emerging economy. The controlling shareholder in family-owned firms transfers resources through pyramidal structure. This expropriation of resources by controlling shareholder can have adverse consequence for the minority shareholders and for the economy as it reduces transparency. Accounting figures are manipulated and there is difficulty in evaluating the actual performance of the firms. Related party transactions are one of the factors used by controlling shareholders to exploit minority shareholder interests. This research highlights the exploitation issues of minority shareholders through related party transactions.

2.2 Theoretical Justification and Research Hypotheses

This study examines the impact of corporate governance variables, namely, independent director index, family directorship, and family ownership on the types of RPTs.

2.2.1 Independent Director Index and RPTs

The concept of "board independence" has been commonly used in many CG reforms. The appointment of INEDs, who are independent from management, is seen as a powerful tool to restrict resource diversion by controlling shareholders. Increasing the independence of corporate directors is one of the main focus of CG reforms. The attributes of INEDs, their composition, financial expertise, and tenure may influence their independence and supervisory roles.

2.2.1.1 Financial Expertise of Independent Directors

Beasley (1996) and Marrakchi, Chtourou, Bedard, and Courteau (2001) have suggested that financial expertise among INEDs is associated with effective board monitoring. As audit committee members, INEDs must be equipped with an accounting background. An accounting-specific expertise is suggested to be crucial for audit committee members given their numerous responsibilities requiring relatively high degrees of accounting sophistication (DeFond, Hann, & Hu, 2005). Studies have indicated that the market reacts positively to the appointment of accounting financial experts to the audit committee, which suggests that INEDs with accounting knowledge improve the said committee's ability to ensure high quality financial information (DeFond et al., 2005).

Prior studies have revealed that the presence of INEDs with financial expertise may enhance the quality of financial reporting process. For

example, such expertise on boards reduces the likelihood of fraud and earnings restatements, promotes the effective mitigation of earnings management, and minimises the likelihood of being associated with the occurrence of internal control problems (Agrawal & Chadha, 2005; Carcello, Hollingsworth, Klein, & Neal, 2006; Krishnan, 2005). Firms with financial reporting problems are unlikely to include financial experts on their audit committees (McMullen, Raghunandan, & Rama, 1996). Other studies have investigated whether the board's financial expertise exerts a positive influence on a firm's financial reporting quality. One research found that the fraction of audit committee members with expertise in accounting or financial management is positively related to financial reporting quality (Felo, Krishnamurthy, & Solieri, 2003).

2.2.1.2 Tenure of Independent Directors

The US Senate report on Enron (US Senate, 2002) revealed that board tenure is another shortcoming of CG practices. Some of Enron's directors served on the board for at least 10 years. More recent trends show that a growing number of companies adopted tenure-related guidelines for INEDs. For example, Hong Kong, Malaysia, Singapore, South Africa, and the United Kingdom recommend a maximum tenure of nine years for INEDs. In Malaysia and United Kingdom, directors with more than nine years of tenure are deemed non-independent unless the company can explain otherwise.

Vafeas (2003) opined that senior directors tend to make decisions that favour management and that CEOs tend to receive high compensation when the committee consists of senior directors (Vafeas, 2003). This was confirmed by Rickling (2014) who revealed that audit committee director tenure is positively associated with the likelihood of a firm repeatedly holding meetings or beating analysts' forecasts. Thus, Rickling (2014) supported the proposal to limit the tenure of directors. Similarly, Chen and Jaggi (2000), Cheng and Courtenay (2006), Morris and Gray (2007) and Morris, Susilowati, and Gray (2012) found a positive association between the ratio of independent directors and corporate disclosures.

In contrast, Liu and Sun (2010) showed a negative relationship between the proportion of long-tenured directors and earnings management, thereby supporting the hypothesis on expertise. Given the two conflicting arguments on the effect of long-tenured directors, this study proposes the INEDs' tenure may have an influence potentially abusive RPTs. Eng and Mak (2003), Barako, Hancock, and Izan (2006) and Nelson, Gallery, and Percy (2010) found a negative relationship between the ratio of outside directors and the firm's voluntary disclosures.

Other studies showed mixed results. Gallery, Gallery, and Supranowicz (2008) revealed a negative relationship between board independence and related party payment, pointing to the monitoring role of independent directors in checking payments to related parties. Lo and Wong (2011) found that firms with a considerable percentage of independent directors voluntarily disclosed the method of transfer pricing of their RPTs. Specifically, firms with a high ratio of independent directors revealed mandatory information on RPTs disciplined by stock market regulatory bodies. Thus, they proposed that independent boards promote more effective monitoring of firm disclosures. In the context of Pakistan, Abdullah, Shah, Gohar, and Iqbal (2011) ascertained that concentrated ownership companies with independent directors have a positive influence on firm performance. Similarly, Khan and Awan (2012) found that incorporating independent directors on the board positively affects firm performance. This finding was supported by Javaid and Saboor (2015) who showed that independent directors positively influence firm performance.

Based on the above discussion, this study argues that independent director index has significant monitoring role on the relationship between RPTs and firm performance. This leads to the following hypothesis for independent director index that improves company performance.

H1a: The independent director index has effect on firm performance that limits RPTb.

H1b: The independent director index has effect on firm performance that limits RPTe.

H1c: The independent director index has effect on firm performance that limits RPTo.

2.2.2 Family directorship and RPTs

Family directorship can affect the interests of minority shareholders because family-owned firms protect a firm from the probability of a hostile take-over, (Gomez-Mejia, Larraza-Kintana, & Makri, 2003). There is a dearth of studies on the performance of family directorship in developing countries, such as Pakistan, where the board is more influential in family-owned firms, which is most common among listed companies. Nicholls and Ahmed (1995) found r appointment of directors with qualification would enhance performance of family-owned firms. La Porta et al. (1999) observed that national institutions failed to protect investor rights in family-owned firms. Claessens et al. (2000) reported that family-owned firms have different levels of controlling and cash flow rights through their pyramidal

ownership. Barontini and Caprio (2006) explored the relationship between ownership structure and firm performance and found that director appointed by family-owned firms is positively related to firm value and operating performance. Similarly, Carney and Gedajlovic (2002), Chang (2003) and Joh (2003) provided evidence that directorship of family-owned firms is significantly related to better performance. Morck, Shleifer, and Vishny (1988) found a negative association between the effects of directorship of family-owned firms and firm performance. However, Filatotchev, Lien, and Piesse (2005) found that directorship of familyowned firms is not related to firm performance. Based on the above discussion, this study argues that family directorship has significant monitoring role on

The relationship between RPTs and firm performance. This leads to following hypothesis that family directorship in Pakistani family-owned firms improves company performance.

H2a: The high proportion of family directorship on the board leads to high RPTb.

H2b: The high proportion of family directorship on the board leads to high RPTe.

H2c: The high proportion of family directorship on the board leads to high RPTo.

2.2.3 Family ownership and RPTs

Jensen and Meckling (1976) concluded that managerial ownership is negatively associated with agency cost, but positively associated with firm performance. This finding is supported by the conflict of interest hypothesis (Gordon et al., 2004). Shleifer and Vishny (1986) proposed that high concentration ownership indicated better monitoring and performance especially when ownership is concentrated in institutional rather than individual investors. Therefore, institutional ownership could enhance firm performance. McConaughy, Walker, Henderson, and Mishra (1998) and Anderson and Reeb (2003) suggested that family-owned firms improve the value of firms. Demsetz and Lehn (1985) found that family-owned firms appointed persons who are closely related to the value of the firm and who could closely monitor management efficiently while reducing problems associated with firms. Maury (2006) reported that family-owned firms improved firm profitability, and the legal environment protected the interest of minority shareholders. Ben- Amar and André (2006) found that family-owned firms that often exert control over voting rights have a small ratio of cash flow rights. Klein, Shapiro, and Young (2005) attested that the relation of performance varies due to the variation of ownership concentration across countries. Villalonga and Amit (2006) posited that highly concentrated family-owned firms with CEO appointed from the family create value for firms when management is under family control.

Previous studies have provided mixed results on the relationship between family ownership/concentration and performance. Demsetz and Lehn (1985) found that family-concentrated firms reduce managerial cost. However, Fama and Jensen (1985) provided evidence that managerial costs are not reduced with the concentration of family-owned firm. Hill and Snell (1988, 1989), Agrawal and Mandelker (1990), Xu and Wang (1997), and Wu and Cui (2002) found positive and significant relations between ownership concentration and accounting profits and firm performance. However, Leech and Leahy (1991), Mudambi and Nicosia (1998), Lehmann and Weigand (2000) and Chen and Cheung (2000) found negative and significant effect of ownership concentration on firm value. However, Prowse (1992) reported that ownership concentration and profitability are not related. Other studies have explored the nonlinear relationship between ownership concentration and firm performance. Thomsen and Pedersen (2000) found that as concentration of ownership increases, firm performance initially improves, but eventually declines. Hence, the value of concentrated ownership is offset by the negative effects of high ownership concentration. Porta et al. (1999) found that expropriation of resources mainly occurs because controlling shareholders have control rights significantly higher than cash flow rights. Claessens et al. (2002), Joh (2003) and Baek, Kang, and Park (2004) agreed that firm value increases with cash flow rights of controlling shareholder and vice versa. Lins (2003) found low firm values are related when control rights are higher than cash flow rights, but this control is insufficient to offset the benefits of concentrated ownership. However, Sánchez- Ballesta and García- Meca (2007) indicated that this relationship of cash flow and control rights is moderated and strong, which would support the argument that ownership is positively associated with firm performance in countries with low investor protection. In the context of Pakistan, Javid and Iqbal (2008) showed a positive effect of ownership concentration on firm performance and negative association between concentration of ownership and corporate governance practices such as disclosures and transparency. This finding is supported by Ali et al. (2015), who revealed a positive effect of ownership concentration on firm value. Abdullah et al. (2011) reported that firms with concentrated ownership structure are negatively related to firm performance. This finding is supported by Irshad, Hashmi, Kausar, and Nazir (2015), who noted that concentrated ownership structure

has a negative relationship with firm performance. Therefore, ownership concentration has a significant monitoring role on the relationship between RPTs and firm performance. This leads to the following hypotheses regarding the concentration of family structure, which moderates the relationship between the related party transaction performances of a company.

H3a: The presence of higher concentration of family ownership increases RPTb.

H3b: The presence of higher concentration of family ownership increases RPTe.

H3c: The presence of higher concentration of family ownership increases high RPTo

3. Methodology

3.1 Data and Sample

The sample size for the current study consists of 108 family-owned firms listed on the KSE (Javid & Iqbal, 2008). These firms are included in the study as they have high market capitalisation. The study used quantitative approach and examined secondary data as well as utilised 1,188 observations between 2004 and 2014 post implementation of corporate governance codes in 2002. Data was obtained from documents, surveys, annual reports, analyst reports, and various studies on family-owned firms in Pakistan.

3.2 Model Specification and Variables

Based on the theoretical justification, the regression models with nonlinear relation were used to examine the relationship between response variables (RPTb, RPTe, and RPTo) and explanatory variables (corporate governance).

Model 1

 $RPTb_{it} = \beta_0 + \beta_1 IDI_{it} + \beta_2 FD_{it} + \beta_3 FO_{it} + \beta_4 FS_{it} + \beta_5 PM_{it} + \varepsilon_{it}$

Model 2 $RPTe_{it} = \beta_0 + \beta_1 IDI_{it} + \beta_2 FD_{it} + \beta_3 FO_{it} + \beta_4 FS_{it} + \beta_5 PM_{it} + \varepsilon_{it}$

Model 3

 $RPTo_{it} = \beta_0 + \beta_1 IDIs_{it} + \beta_2 FD_{it} + \beta_3 FO_{it} + \beta_4 FS_{it} + \beta_5 PM_{it} + \varepsilon_{it}$

Where,

 $RPTb_{it}$ amount of RPT benefit-based that is likely to result in expropriation at year t

 $RPTe_{ii}$: amount of RPT expense-based that is likely to result in expropriation at year t

 $RPTo_{it}$: amount of RPT other-based that is likely to result in expropriation at year t

The independent variables are:

 IDI_{it} : Independent director Index of independent directors in the firm at year t.

 FD_{it} : natural log of the shareholding amount by a family member as a director in the Board of Director (BoD) in i firm at year t

FO_{it}: family concentration of major shareholder in the firm at year t (%)

FS it: firm size of firm i at year t

PM it: profitability of firm i at year t

ε: standard error

3.2 Variables

The dependent variables are the RPTs, namely RPTb, RPTe, and RPTo. The RPTb is measured by the total amount of bonus shares, convertibles, and rights to the related party as a measure of RPTs. The RPTe is measured using the total dollar amount of organisational, insurance, and royalty expenses, as well as other expenses to related party is a measure of RPTs. The RPTo is measured by utilising the total dollar amount of ordinary shares, dividends, donations, interests, investments, purchases of assets, sales of asset, employee benefits, leases and loans, and advances to related party as a measure of RPTs. The independent variables of the study are corporate governance variables, namely IDI, FD and FO. The IDI includes three dimensions for the measurement of the autonomy of independent directors of family-owned firms. Figure 1 shows that overall, IDI is computed as a weighted sum of three sub-dimensions, the composition of independent directors (IDC), the financial expertise of independent directors (IDFC), and the tenure of the independent directors (IDT). The first step in the calculations involve constructing all sub-indices separately by assigning a specific weight to each dimension using the principal component method (Filmer & Pritchett, 2001; Harttgen & Klasen, 2012; Sahn & Stifel, 2003; Sahn & Stifel, 2000), Javid and Iqbal (2010b) of the

CG index development and Khan and Yusof (2017) of the terrorist economic impact evaluation model (TEIE Model) development. To calculate the sub-dimensions, the max-min approach of the United Nations Development Program is adopted. The second step is to take the mean of the calculated dimensions to obtain the overall IDI for a particular year. Family directorship is computed as the percentage of shares held by family members on the board. Family ownership is measured in terms of the percentage of total equity held by each controlling shareholder (Demsetz & Lehn, 1985; Gul, Kim, & Qiu, 2010; Maury, 2006; Wruck, 1989).

In addition, few control variables were used, namely firm size and profit margin. Firm size refers to the value of the total assets of a firm. Considering firm level variables, the size of firm is kept as control variable and inverse relationship between ownership concentration and firm size, which is expected for risk averting and risk neutral effects. In larger firms, the stake of ownership is greater, and higher price of shares would reduce degrees of concentration. These factors are important determinants in assessing firm performance and propagated by many theorists in business literature.

Kajola (2008) used variable net profit margin to represent the performance variable concerned with firm operations. This ratio is especially important because it links core business operations with the profit the business generates. At the end of a fiscal year, the net profit margin ratio indicates how well a firm transformed its business activities into retained earnings. The net profit margin is ideally calculated by dividing the net profit of the firm by its sales revenue for the year. Therefore, this ratio describes the profit sales relationship, an important element for measuring firm performance.

3.3 Estimation techniques

The study conducted content analysis to categorise RPTs into three types, namely RPTb, RPTe, and RPTo. The study used panel data analysis through statistical software, namely, Stata version 14. Generalized Method of Moments (GMM) estimator is employed to empirically test the developed hypotheses in this study. GMM estimators are used for robustness testing and control of heteroscedasticity problems in the data. Researchers such as Arellano and Bond (1991), Arellano and Bover (1995) and Blundell and Bond (1998) have developed GMM estimators applied to panel data in following situations: First, such estimators are applied when the data involved a short period and a large number of observations (Roodman, 2006a, 2006b, 2009); second, when a linear relationship exists; third, in cases involving an estimation where the dependent variable is dynamic, meaning that its current value depends on its values in previous

time periods; fourth, when independent variables are not strictly exogenous; and finally, when the panel data displays heteroskedasticity and autocorrelation within individuals but not across them (Roodman, 2009). The GMM method is an instrumental variable approach. The instruments include all variables used in the estimation (and previously employed in the OLS regression). One of the main advantages of this method is its usage to avoid autocorrelation and heteroskedasticity problems, which often plague the standard OLS method and in turn affects estimation efficiency (Baum, Schaffer, & Stillman, 2003). The heteroskedasticity issue refers to the circumstance wherein the conditional variance of variable X is not constant with that of variable Y. The classical linear regression model assumes that each variable has the same variance or constant (i.e., homoscedastic). Another notable advantage of the GMM method is its provision of a unified framework when analysing results of other common estimation method such as the OLS and the instrumental variables (IV) approach (Kennedy, 2008). The GMM estimator can be identified by including the exact number of instruments as the number of independent variables. The model for this study is similar to that used by Arellano and Bond (1991), Xu and Wang (1999), Anderson and Reeb (2003), and Claessens et al. (2002) which ignored non-linearity to keep model parsimony and to prevent significant multicollinearity and autocorrelation issues (Gujarati & Porter, 2009). Raw data was examined to produce descriptive statistics.

The potential problem of endogeneity exists in empirical studies that focus on the relationship between ownership concentration and firm value (Andres, 2008). A reverse causality relationship may occur between ownership and firm value. The controlling shareholder keeps their shares in a well-performing firm while they hand over control in a firm with poor performance. Moreover, controlling shareholders have a high membership in the board of directors, which enables the controlling shareholder to acquire increased information about forecasting future firm performance. Therefore, firm value could be determined by the ownership concentration of the controlling shareholder (Andres, 2008). However, firm performance is claimed to determine ownership concentration for several reasons. Although major shareholders have advantageous information regarding the future prospects of firms, presuming that they can forecast the firm performance over the decades appears irrational (De Andres & Vallelado, 2008). Thus, the endogeneity test is conducted to examine the existence of reverse causality. The study applies an augmented regression test, the Durbin-Wu-Hausman (DWH) proposed by Davidson and MacKinnon (1993) to check for endogeneity issues. The test follows a two-step procedure. First, the potential endogenous variable is regressed on all the exogenous variables in the system, and the residuals are calculated. Second, the residuals are used in place of the endogenous variable in the original

model. If the coefficient on residual is significant, then the variable is endogenous. The potential candidates for endogeneity in this study are RPTb, RPTe, and RPTo. Results from the DWH test indicate that these variables are exogenous.

4. Results and Discussions

Table 1 shows descriptive statistics of RPTb, RPTe, RPTo, firm size, profitability, and corporate governance variables, namely INEDs, family directorship, and family ownership. The mean of the variables determines the overall value of the variables across all family-owned firms in the KSElisted sample. The mean RPTs determines the number of RPTs of familyowned firms listed on the KSE. All RPTs are categorised into three types. The first category is RPTb, which has a mean value of 13.71. The second category is RPTe, which has a mean value of 15.99. The third category is RPTo, which has a mean value of 17.21. The mean board composition determines the number of independent and INEDs of firms. The IDI, which shows the percentage of independent directors in the board, has a mean value of 23.95%. This value can be compared to that of Gul (2012) at 15.54%. Family ownership indicated by family-owned firms is 54.20%. Family directorship determines that on average, every firm has around 31.44%. The average log of assets, i.e., firm size is 14.80 in the sample. The average ratio of profit margin is 12.46% in the sample. Standard deviation compares the overall deviation or divergence prevalent in the data of the sample. This variation determines the diversity and different patterns of family-owned firms included in the sample. The least amount of deviation is observed in the return on assets, whereas the most deviation is noted in the board size. The confidence level is at 95%.

Table1: Descriptive statistics						
Variable	Mean	Std. Dev.	Min	Max		
RPTb	14.12	0.51	12.61	15.04		
RPTe	16.42	1.30	13.39	19.94		
RPTo	17.70	1.27	14.14	20.04		
IDI _{it}	23.95	9.43	0.00	80.30		
FD	31.44	25.61	1.13	99.94		
FO	54.20	21.55	9.24	98.85		
FS	14.80	1.80	7.85	19.48		
PM	12.46	13.37	1.00	58.43		

Note: This table shows descriptive statistics of variables; where, RPTb = RPT benefitbased transactions; RPTe = RPT expense-based transactions; RPTo = RPT other-based transactions; IDI = Independent director index; FD = family director; FO = family ownership; PM = profit margin; and FS = firm size

Table 2 shows the correlation among the variables as well as the Pearson correlation of the variables used to conduct this research. Pearson correlation aims to measure the extent of multicollinearity among variables. In Table 5.2, the variables are compared horizontally and diagonally to determine their correlation. The relationship among RPTs (RPTb, RPTe, and RPTo) and CG variables (IDI and FO) shows that IDI and FO are negatively correlated with RPTb, RPTe and RPTo. However, FD is positively correlated with RPTb and RPTo. Similarly, FD is negatively correlated with RPTe. Finally, RPTb, RPTe and RPTo are positively correlated with PM and Firm size. The last column in Table 5.2 shows the VIF, c indicating multicollinearity issue in the research model. The highest value of VIF is 1.62, which suggests that multicollinearity is not significant in this research. Although there is some significant correlation among the variables, the value of multicollinearity should not exceed 0.8 to be significant; hence, the issue of multicollinearity can be ignored in this scenario (Gujrati, 1992).

Table 2: Correlation Matrix										
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	VIF	
(1) RPTb	1								1.62	
(2) RPTe	0.395	1							1.25	
(3) RPTo	0.453	0.136	1						1.24	
(4) IDIs	-0.040	-0.040	-0.030	1					1.16	
(5) FD	0.114	-0.055	0.055	-0.112	1				1.18	
(6) FO	-0.031	-0.165	-0.006	-0.019	0.081	1			1.07	
(7) PM	0.068	0.007	0.001	0.062	-0.199	-0.147	1		1.05	
(8) FS	0.214	0.171	0.14	-0.005	-0.309	-0.244	0.180	1	1.11	

Table 2: Correlation Matrix

Note: This table shows the correlation among variables; where, RPTb = RPT benefitbased transactions; RPTe = RPT expense-based transactions; RPTo = RPT other-based transactions; IDI = independent director index; FD = family director; FO = family ownership; PM = profit margin; FS = firm size

Table 3 shows the regression results of two-step system GMM using Model 1, where the dependent variable is RPTb, and independent variables are INEDs, family directorship, and family ownership. The control variables are profit margin and firm size. The table shows significant results in which expropriation of resources occur in family-owned firms through benefit-based RPTs. Table 6 in Appendix B details of the two-step system GMM regression results. Columns 1 to 7 show the addition and elimination of independent variables with respect to dependent variable, RPTb. Column 1 of Table 3 further shows that IDI is negatively related to RPTb, whereas family directorship and family ownership are positively related to RPTb. Furthermore, family director is insignificant to RPTb as shown in Column 1. Family director becomes positively significant by dropping variables, such as IDI, as shown in Column 2. This positive relationship of family director is consistent with the findings of Barontini and Caprio (2006), who found that directors appointed by family-owned firms are positively related to firm value and operating performance. Similarly, Carney and Gedajlovic (2002), Chang (2003) and Joh (2003) found empirical evidence that the directorship of family-owned firms is significantly related to enhanced performance. This finding is consistent with the impact of family directorship that is positively related with RPTb. The negative relationship with independency of board is consistent with the finding of Gallery et al. (2008), who claimed the negative relationship between independence of board and related party payment and the monitoring role of independent directors in checking payments to related to parties. The positive relationship of the board with RPTb is consistent with the conclusions by Carney and Gedailovic (2002) and Joh (2003) who

found that the directorship of family-owned firms is significantly related to its enhanced performance.

Table 3 shows that AR (1) has significant values which indicates the null hypothesis of autocorrelation among error term in first difference is rejected. AR (2) is insignificant and shows that error terms in level regressions are not correlated. The results of AR (1) and AR (2) of Table 3 show that GMM is correctly specified and no identification issues are present.

	(1)	(2)
Related Party Transactions-b (t-1)	0.199***	0.199***
Independent Non-Executive Directors(IDI)	(0.02) -0.029*** (0.007)	(0.019)
Family Directorship	0.000 (0.000)	0.001** (0.000)
Family Ownership	0.006***	0.005***
Profit Margin	(0.001) 0.001*	(0.001) 0.001**
-	(0.000)	(0.000)
Firm Size	0.007 (0.004)	0.009* (0.005)
Number of Groups	108	108
Observations	1,188	1,188
AR(1)	-	0.007
AR(2)	-	0.707

Table 3: Corporate Governance and Related Party Transactions (b)

Note: This table shows the regression of model 1 by using two-step GMM; where dependent variable is RPTb and independent variables are independent director index, family directorship, family ownership and control variables are profit margin and firm size. *p < 0.1; **p < 0.05; ***p < 0.01.

Table 4 shows the regression results of the two-step system GMM using Model 2, where the dependent variable is RPTe. Independent variables include INEDs, family directorship, and family ownership. The control variables are profit margin and firm size. The results indicate that expropriation of resources occurred in family-owned firms through expense-based RPTs. Table 7 of Appendix B presents the details of the two-step system GMM regression results. Columns 1 to 7, with respect to dependent variables. Column 1 of Table 4 further shows that IDI and family ownership are negatively related to RPTe. Family directorship is

negatively related to RPTe, but this relationship is insignificant. Family director becomes negatively significant by dropping variables, such as IDI, as shown in Column 2 of Table 4. The negative relationship of family director with respect to RPTe is consistent with the findings of Morck et al. (1988) who reported the negative association between effects of directorship of family-owned firms and firm performance. The negative relationship of family ownership with respect to RPTe is consistent with the findings of Lauterbach and Tolkowsky (2007) and Amzaleg and Barak (2011) who found a negative relationship between RPT and concentration of family ownership of Israeli firms. Table 4 further shows the significant values of AR (1) and the rejection of the null hypothesis of autocorrelation among error terms in the first difference. AR (2) is insignificant, showing that error terms in level of regressions are not correlated. The results of AR (1) and AR (2) in Table 4 show that GMM is correctly specified and no identifications issues emerged.

	(1)	(2)
Related Party Transactions-e (t-1)	0.391***	0.396***
	(0.020)	(0.017)
Independent Non-Executive Directors (IDI)	-0.005*	-
	(0.001)	-
Family Directorship	-0.003	-0.004**
	(0.002)	(0.002)
Family Ownership	-0.029***	-0.026***
	(0.003)	(0.004)
Profit Margin	0.007***	-
	(0.002)	-
Firm Size	0.029	-
	(0.027)	-
Number of Groups	108	108
Observations	1,080	1,080
AR(1)	-	.030
AD(2)		559

Table 4: Corporate Governance and Related Party Transactions (e)

Note: This table shows the regression of Model 2 using two-step GMM; where dependent variable is RPTe and independent variables are independent director index, family directorship, family ownership; and control variables are profit margin and firm size. *p < 0.1; **p < 0.05; ***p < 0.01.

Table 5 shows the regression results of the two-step system GMM using Model 3, where the dependent variable is RPTo. Independent variables are IDI, family directorship, and family ownership. The control variables are profit margin and firm size. Results show expropriation of resources occurs in family-owned firms through other-based RPTs. Table 8 of Appendix B presents the details of the two-step system GMM regression results. Columns 1 to 7, with respect to dependent variable, RPTo, show the addition and elimination of independent variables. Column 1 of Table 5 further shows that IDI and family director are negatively related to RPTo. Family ownership is positively related to RPTo. The IDIs become more significant by dropping family directorship and firm size as shown in Column 2. The significant negative relationship of having independent directors in the board with RPTo is consistent with the findings of Fama (1980), Fama and Jensen (1983), (Gordon & Henry, 2005) and Gao and Kling (2008). Table 5 further shows the significant values of AR(1) and the rejection of the null hypothesis of autocorrelation among error terms in the first difference. AR(2) is insignificant and shows that error terms in level regressions are not correlated. The results of AR(1) and AR(2) indicate that GMM is correctly specified and no identifications issues emerged.

Tuble 5. Corporate Governance and Related Fairly Transactions (6)						
	(1)	(2)				
Related Party Transactions-0 (t-1)	0.411***	0.409***				
	(0.026)	(0.021)				
Independent Non-Executive Directors (IDI)	-0.004*	-0.004**				
-	(0.001)	(0.001)				
Family Directorship	-0.007***	-				
	(0.002)	-				
Family Ownership	0.042***	0.043***				
	(0.002)	(0.002)				
Profit Margin	0.009***	0.009^{***}				
	(0.001)	(0.001)				
Firm Size	-0.023*	-				
	(0.012)	-				
Number of Groups	108	108				
Observations	1,080	1,080				
AR(1)	0.035	-				
AR(2)	0.854	-				

 Table 5: Corporate Governance and Related Party Transactions (o)

Note: This table shows the regression of Model 3 by using the two-step GMM, where dependent variable is RPTo and independent variables are independent director index, family directorship, family ownership and control variables are profit margin and firm size. *p < 0.1; **p < 0.05; ***p < 0.01.

5. Conclusion and Implications

This is an empirical study on the impact of independent director index, family directorship, and concentration of ownership on types of RPTs that

prevail in family-owned firms in developing countries such as Pakistan. It analysed the panel data of 108 family-owned firms listed on the KSE from 2004 to 2014. Different panel least squares models were employed to examine the relationship between different types of RPTs (RPTb, RPTe, and RPTo) and corporate governance variables (INEDS, family directorship, and family ownership). Findings show that corporate governance variables, particularly independent director index, have a negative relationship with related party transactions. This finding pointed to weaker corporate governance in family-owned firms, where major shareholders expropriated resources through RPTs, namely, RPTb and RPTe. The study also found that IDI (independent directors index) have a negative relationship with RPTs, as most family-owned firms have low independence, as most decisions are taken by major shareholders, and the interest of minority shareholder is compromised. This finding is supported by IFC (2007). Corporate governance in Pakistan in short indicates low or poor protection of minority shareholders. Further, family directors are negatively related to RPTs, as most of shareholders are family directors and the concentration of ownership structure of family-owned firms is negatively related with RPTs, and this concentration has a negative tendency for the resource expropriation of family-owned firms. The negative relationships of corporate governance and concentration of major shareholder will lead to exploitation of the interest of minority shareholders. These results are consistent with Ikram and Naqvi (2005), (Ibrahim, 2006) and Mehboob et al. (2015). They indicated that corporate governance and ownership concentration negatively impact the interest of minority shareholders due to expropriation of resources through certain RPTs.

The study also focused on regulatory authority's role to on ensuring disclosure requirements are adhered to by companies in increasing transparency. (the study was conducted after the implementation of the CG Codes of the SECP in 2002). The results also direct the attention to the role of SECP and the significance of having INEDs on boards by developing IDI in three dimensions, i.e., composition, financial expertise, and tenure. The board size of family-owned firms shows a low proportion of INEDs. Therefore, SECP must increase autonomy of the Board by selecting directors from outside the firm. These independent directors should have financial expertise to mitigate transfer-pricing policy and all RPTs that are not priced toward the advantage of major shareholders. Family-owned firms that do not adhere to the CG codes should be strictly dealt with. Therefore, the CG code should be reviewed during the tenure of independent directors to ensure that its implementation remains true to its letter and spirit. This could enhance transparency and increase the confidence of minority shareholders. The SECP and all stock exchanges should be required to improve their regulations to ensure that the financial statements of firms for the last five years are easily accessible from their websites and may be found in the archives of respective stock exchanges. The SECP and the KSE should conduct a survey to come up with the official and authentic list and ranking of family-owned firms each year according to the cash flow rights of the controlling family. This survey will improve data repository for investors and researchers and enhance CG. Researchers, scholars, and analysts should be encouraged and guided to continue further research to determine the other aspects of RPTs. In sum, this study has suggested measures that should be adopted by minority shareholders to restrict the expropriation by major shareholders.

	mppenaix m 1	pes or	In The ported by fulling owned minis				
		1	Bonus shares issued				
		2	Bonus shares received during the period (No. of				
			Shares)				
	Types of RPT	3	Bonus and earned leaves				
	benefit based	4	Bonus shares issued				
	transactions	5	Bonus shares issued at nominal value				
1	(Bonus	6	Bonus shares received at nominal value				
	Convertible and	7	Bonus units received during the period (No. of				
	Right)		Units)				
		8	Convertible preference shares received (No. of				
			shares)				
		9	Shares received against right subscription (No.				
			of Shares)				
		1	Organizational expense recovered				
	Types of RPT	2	Organizational expense paid				
	Expense based	3	Insurance claims received				
transactions		4	Insurance premium paid				
	(Organizational	5	Insurance refund / cancellation				
	Insurance	6	Insurance services				
2	expense,	7	Premium due but unpaid				
	Royalty	8	Premium written				
	expense, Other	9	Royalty and technical fee				
	Expenses)	10	Royalty expense				
		11	Expenses incurred by subsidiary company on				
			Company's behalf				
		12	Expenses incurred on behalf of subsidiary				
			company				
		13	Reimbursement of expenses to the Company				
		14	Reimbursement of expenses by the Company				

Appendix A: Types of RPTs reported by family-owned firms

	Appendix A: (Continued)								
		15	Reimbursement of expenses to associated						
			companies						
		16	Reimbursement of expenses to the Company						
		17	Maintenance and utility charges						
		18	Payment of utility charges						
		1	Redemption of units (No. of Units)						
		2	Gain on redemption of units						
		3	Capital gain earned						
		4	Income on sale and subsequent purchase of shares						
		5	Share deposit money received						
	Other RPT	6	Shares received against splitting of share capital						
3	(Ordinary	7	Shares received against subscription of right						
	Share) shares								
		8	Units issued on conversion to open end fund						
		9	Units purchased during the period (No. of Units)						
		10	Redemption of units						
		11	Sale and purchase of Securities						
		12	Subscription of ordinary shares						
		1	Dividend income						
		2	Dividend income Contributions to Staff Provident						
			Fund						
	Other BPT	3	Ordinary dividend paid						
4	(Dividend)	4	Dividend Paid						
	(Dividend)	5	Dividend received						
		6	Limited Dividend income						
		7	Ordinary dividend paid						
		8	Ordinary dividend received						
		9	Preference dividend paid						
5	Other RPT	1	Donation Payable						
5	(Donation)	2	Donation paid						
		1	Interest income on bridge financing						
		2	Interest income on subordinated loan						
		3	Interest / mark-up paid						
		4	Interest income						
6	Other DDT	5	Interest Income earned on advance						
0	(Interact)	6	Interest on long term loans to executives						
	(Interest)	7	interest on short term loan						
		8	Interest paid						
		9	Interest received on bank accounts						
		10	Mark up income earned						
		11	1 Mark up on advances given						

Appendix A: (Continued)

	Appendix A: (Continued)					
		1	Investment made			
		2	Disposal of investment			
		3	Disinvestments during the period			
		4	Investment - available for sale			
		5	Investment held			
	Other DDT	6	Investment in convertible preference right shares			
7	(Investment)	7	Investment in issue of right shares			
	(Investment)	8	Investment in right shares			
		9	Investment in shares outstanding			
		1	Purchase of goods			
		2	Purchase of an intangible asset			
		3	Purchase of fixed assets			
	Other DDT	4	Purchase of investment Property			
8	(Purchasos	5	Purchase of operating fixed asset			
	(Fulchases	6	Purchase of packing material -			
	OI Assets)	7	Purchase of property, plant and equipment			
		8	Purchase of shares			
		9	Purchase of units			
		10	Purchases of long-term investments			
		1	Disposal of operating assets under Company policy			
		2	Proceeds against redemption of units			
		3	Proceeds from disposal of non - current assets held			
	Other RPT		for sale			
9	(Sale of	4	Sale of fixed assets			
	asset)	5	Sale of goods and services			
		6	Sale of government securities			
		7	Sale of packing material -			
		8	Sale of property, plant and equipment			
		9	Sale proceeds of long-term investments			
		1	Post-Employment benefit plan			
		2	Contribution by the Company			
	Other RPT	3	Contribution paid during the year			
10	(Employee	4	Contribution to employees' provident fund			
	benefit)	5	Gratuity funds			
		6	Salaries and other short-term employees' benefits			
		7	Remuneration and other benefits			
11	Other RPT	1	Lease income			
11	(Lease)	2	Lease rentals received			

Annondiv A. (Continued)

	Appendix A. (Continued)				
		1	Loan / advance recovered		
		2	Loan and advances repaid from executives		
		3	Loans and advances disbursed during the year		
		4	Repayment of short term loans by the subsidiary		
			company		
	Other RPT	5	advance adjusted against shares issued		
12	(Loan and	6	Advance against equity		
	advance)	7	Advance against investment		
		8	Advance against purchase of fixed assets		
		9	Advance for purchase of land		
		10	Long term loan recovered		
		11	Loan adjusted against equity		

Appendix A: (Continued)

Appendix B

Table 6: Corporate Governance and Related Party Transactions (b)								
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	
Related Party								
Transactions-b (t-1)	0.199***	0.201***	0.204***	0.196***	0.211***	0.179***	0.199***	
	(0.020)	(0.019)	(0.020)	(0.020)	(0.020)	(0.019)	(0.019)	
Independent director								
index (IDI)	-0.029***	-0.030***	-0.031***	0.005***			-	
	(0.007)	(0.007)	(0.008)	-				
Family Directorship								
(FD)	0.000	0.000	0.000	0.000	0.000	0.001*	0.001**	
	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	
Family Ownership								
(FO)	0.006***	0.006***	0.006***	0.006***	0.005***	0.006***	0.005***	
	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	
Profit Margin (PM)	0.001*	0.001*	0.001*	0.001**	0.001**		0.001**	
	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)		(0.000)	
Firm size (FS)	0.007						0.009*	
	(0.004)						(0.005)	
Observations	1,080	1,080	1,080	1,080	1,080	1,080	1,080	

Note: This table shows the regression of model 1 by using two step GMM where dependent variable is rptb and independent variables are IDI, FD and FO and control variables are Profit Margin and Firm size. * p<0.1; ** p<0.05; *** p<0.01.

Appendix B:	(Continued)
--------------------	-------------

	(1)	(2)	(3)	(4)	(5)	(6)	(7)
Related Party							
Transactions-e (t-1)	0.391***	0.393***	0.398***	0.399***	0.401***	0.392***	0.396***
	(0.020)	(0.020)	(0.021)	(0.020)	(0.018)	(0.017)	(0.017)
Independent director							
index (IDI)	-0.005*	-0.005*	-0.005**	-0.006***	-0.006***	-0.006***	-
	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	-
Family Directorship							
(FD)	-0.003	-0.003	-0.003	-0.003	-0.003	-0.003*	-0.004**
	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)
Family Ownership							
(FO)	-0.029***	-0.027***	-0.029***	-0.031***	-	-	-0.026***
	(0.003)	(0.004)	(0.004)	(0.004)	-	-	(0.004)
Profit Margin (PM)	0.007***	0.007***	0.007***	0.006***	0.006***	-	-
	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	-	-
Firm size (FS)	0.029	-	-	-	-	-	-
	(0.027)	-	-	-	-	-	-
Number of Groups	108	108	108	108	108	108	108
Observations	1,080	1,080	1,080	1,080	1,080	1,080	1,080

Table 7: Corporate Governance and Related Party Transactions (e)

Note: This table shows the regression of model 2 by using two step GMM where dependent variable is rpte and independent variables are IDI, FD and FO and control variables are Profit Margin and Firm size. * p<0.1; ** p<0.05; *** p<0.01

	Table 8: Corporate Governance and Related Party Transactions (o)						
	(1)	(2)	(3)	(4)	(5)	(6)	(7)
Related Party							
Transactions-o					0.418**		
(t-1)	0.411***	0.415***	0.416***	0.418***	*	0.404^{***}	0.409***
	(0.026)	(0.026)	(0.026)	(0.025)	(0.027)	(0.027)	(0.021)
Independent							
director					-		
index (IDI)	-0.004*	-0.004*	-0.004*	-0.005**	0.005**	-0.004	-0.004**
	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)
Family					-		
Directorship					0.008**		
(FD)	-0.007***	-0.007***	-0.007***	-0.007***	*	-0.010***	-
	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.001)	-
Family							
Ownership							
(FO)	0.042***	0.041***	0.041***	0.040***	-	-	0.043***
	(0.002)	(0.002)	(0.002)	(0.003)	-	-	(0.002)
Profit					0.009**		
Margin (PM)	0.009***	0.009***	0.009***	0.009***	*	-	0.009***
	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	-	(0.001)
Firm size							
(FS)	-0.023*	-0.026**	-0.026**	-	-	-	-
	(0.012)	(0.011)	(0.011)	-	-	-	-
Number of							
Groups	108	108	108	108	108	108	108
Observations	1,080	1,080	1,080	1,080	1,080	1,080	1,080

Note: This table shows the regression of model 3 by using two step GMM where dependent variable is rpto and independent variables are IDI, FD and FO and control variables are Profit Margin and Firm size. p<0.1; p<0.05; p<0.01.

Appendix C



Figure 1: Independent Director Index (IDI) of Family Owned Firms

Notes

- ^{1.} Where insiders are referred to as major or controlling shareholders, family, financial institutions, or government. Outsider shareholders are often referred to as minority shareholders (La Porta et al., 1998),
- ^{2.} A Survey of Corporate Governance Practices by International Finance Corporation in Pakistan 2007
- ^{3.} (US GAAP Statement of Financial Accounting Standards 57).
- ^{4.} As stated in paragraph 29.2, IAS 24 (revised) (PricewaterhouseCoopers, 2010)

References

- Abbas, A., Naqvi, H. A., & Mirza, H. H. (2013). Impact of large ownership on firm performance: A case of non-financial listed companies of Pakistan. *World Applied Sciences Journal*, 21(8), 1141-1152.
- Abdullah, F., Shah, A., Gohar, R., & Iqbal, A. M. (2011). The effect of group and family ownership on firm performance: Empirical evidence from Pakistan. International Review of Business Research Papers Vol. 7. No. 4. July 2011 Pp. 177-194.
- Afgan, N., Gugler, K., & Kunst, R. (2016). *The effects of ownership* concentration on performance of Pakistani listed companies. Paper presented at the CBU International Conference Proceedings. Retrieved

from CBU International Conference on Innovations In Science and Education MARCH 23-25, 2016, Prague, Czech Republic.

- Afza, T., & Nazir, M. S. (2015). Role of Institutional Shareholders' Activism in Enhancing Firm Performance: The Case of Pakistan. *Global Business Review*, 16(4), 557-570.
- Agrawal, A., & Chadha, S. (2005). Corporate governance and accounting scandals. *The Journal of Law and Economics*, 48(2), 371-406.
- Agrawal, A., & Knoeber, C. R. (2012). Corporate Governance and Firm Performance, Chapter 26 in Thomas, Christopher R. and William F. Shughart II, eds: Oxford Handbook in Managerial Economics.
- Agrawal, A., & Mandelker, G. N. (1990). Large shareholders and the monitoring of managers: The case of antitakeover charter amendments. *Journal of Financial and Quantitative analysis*, 25(2), 143-161.
- Aharony, J., Wang, J., & Yuan, H. (2010). Tunneling as an incentive for earnings management during the IPO process in China. *Journal of Accounting and Public Policy*, 29(1), 1-26.
- Ali, Y., Tahir, S. H., & Nazir, N. (2015). Impact of Ownership Structure on Firm Value: A Quantitative Analysis of All Listed Companies in Karachi Stock Exchange (KSE) Pakistan.
- Ameer, B. (2013). Corporate governance-issues and challenges in Pakistan. International Journal of Academic Research in Business and Social Sciences, 3(4), 79.
- Amzaleg, Y., & Barak, R. (2011). Ownership Concentration and the Value Effect of Related Party Transactions. Available at SSRN: https://ssrn.com/abstract=1959557 or http://dx.doi.org/10.2139/ssrn.1959557.
- Anderson, R. C., & Reeb, D. M. (2003). Founding-family ownership and firm performance: evidence from the S&P 500. *The Journal of Finance*, 58(3), 1301-1328.
- Andres, C. (2008). Large shareholders and firm performance—An empirical examination of founding-family ownership. *Journal of Corporate Finance*, *14*(4), 431-445.
- Arellano, M., & Bond, S. (1991). Some tests of specification for panel data: Monte Carlo evidence and an application to employment equations. *The review of economic studies*, 58(2), 277-297.
- Arellano, M., & Bover, O. (1995). Another look at the instrumental variable estimation of error-components models. *Journal of econometrics*, 68(1), 29-51.
- Azeez, A. (2015). Corporate governance and firm performance: evidence from Sri Lanka. *Journal of Finance*, *3*(1), 180-189.
- Bae, K. H., Kang, J. K., & Kim, J. M. (2002). Tunneling or value added? Evidence from mergers by Korean business groups. *The Journal of Finance*, 57(6), 2695-2740.

- Baek, J.-S., Kang, J.-K., & Park, K. S. (2004). Corporate governance and firm value: Evidence from the Korean financial crisis. *Journal of Financial Economics*, 71(2), 265-313.
- Barako, D. G., Hancock, P., & Izan, H. (2006). Factors influencing voluntary corporate disclosure by Kenyan companies. *Corporate Governance: An International Review*, 14(2), 107-125.
- Barontini, R., & Caprio, L. (2006). The effect of family control on firm value and performance: Evidence from continental Europe. *European Financial Management*, *12*(5), 689-723.
- Baum, C. F., Schaffer, M. E., & Stillman, S. (2003). Instrumental variables and GMM: Estimation and testing. *Stata journal*, *3*(1), 1-31.
- Baysinger, B. D., & Butler, H. N. (1985). Corporate governance and the board of directors: Performance effects of changes in board composition. *Journal of Law, Economics, & Organization, 1*(1), 101-124.
- Beasley, M. S. (1996). An empirical analysis of the relation between the board of director composition and financial statement fraud. *Accounting review*, 443-465.
- Ben-Amar, W., & André, P. (2006). Separation of ownership from control and acquiring firm performance: The case of family ownership in Canada. *Journal of Business Finance & Accounting*, 33(3-4), 517-543.
- Bennedsen, M., & Wolfenzon, D. (1998). The balance of power in close corporations. 1998-15, University of Copenhagen. Department of Economics. Centre for Industrial Economics. Retrieved from (No. 1998-15). University of Copenhagen. Department of Economics. Centre for Industrial Economics Retrieved from
- Berle, A., & Means, G. (1932). Private property and the modern corporation.
- Berle Private property and the modern corporation. New York: Macmillan.
- Bertrand, M., Mehta, P., & Mullainathan, S. (2002). Ferreting out tunnelling: An application to Indian business groups. *The Quarterly Journal of Economics*, 117(1), 121-148.
- Bhutta, A. I., Knif, J., & Sheikh, M. F. (2016). Ownership Concentration, Client Importance, and Earnings Management: Evidence from Pakistani Business Groups. Available at SSRN: https://ssrn.com/abstract=2852255 or http://dx.doi.org/10.2139/ssrn.2852255.
- Blundell, R., & Bond, S. (1998). Initial conditions and moment restrictions in dynamic panel data models. *Journal of econometrics*, 87(1), 115-143.
- Buysschaert, A., Deloof, M., & Jegers, M. (2004). Equity sales in Belgian corporate groups: expropriation of minority shareholders? A clinical study. *Journal of Corporate Finance*, *10*(1), 81-103.

- Cadbury, A. (1992). Report of the committee on the financial aspects of corporate governance (Vol.
- 1): Gee.
- Carcello, J. V., Hollingsworth, C. W., Klein, A., & Neal, T. L. (2006). Audit committee financial expertise, competing corporate governance mechanisms, and earnings management. Available at SSRN: https://ssrn.com/abstract=887512 or http://dx.doi.org/10.2139/ssrn.887512.
- Carney, M., & Gedajlovic, E. (2002). The co-evolution of institutional environments and organizational strategies: The rise of family business groups in the ASEAN region. *Organization studies*, 23(1), 1-29.
- Chang, S. J. (2003). Ownership structure, expropriation, and performance of group-affiliated companies in Korea. Academy of Management Journal, 46(2), 238-253.
- Chang, S. J., & Hong, J. (2000). Economic performance of group-affiliated companies in Korea: Intragroup resource sharing and internal business transactions. *Academy of Management Journal*, *43*(3), 429-448.
- Chen, C. J., & Jaggi, B. (2000). Association between independent nonexecutive directors, family control and financial disclosures in Hong Kong. *Journal of Accounting and Public Policy*, *19*(4-5), 285-310.
- Chen, Z., & Cheung, Y. (2000). 'Corporate governance, firm performance and agency conflicts in closely-held firms: evidence from Hong Kong. working paper, City University of Hong Kong, Hong Kong.
- Cheng, E. C., & Courtenay, S. M. (2006). Board composition, regulatory regime and voluntary disclosure. *The International Journal of Accounting*, 41(3), 262-289.
- Cheung, Y.-L., Qi, Y., Rau, P. R., & Stouraitis, A. (2009). Buy high, sell low: How listed firms price asset transfers in related party transactions. *Journal of Banking & Finance*, *33*(5), 914-924.
- Cheung, Y.-L., Rau, P. R., & Stouraitis, A. (2006). Tunneling, propping, and expropriation: evidence from connected party transactions in Hong Kong. *Journal of Financial Economics*, 82(2), 343-386.
- Claessens, S., Djankov, S., Fan, J. P., & Lang, L. H. (2002). Disentangling the incentive and entrenchment effects of large shareholdings. *The Journal of Finance*, *57*(6), 2741-2771.
- Claessens, S., Djankov, S., & Lang, L. H. (2000). The separation of ownership and control in East Asian corporations. *Journal of Financial Economics*, 58(1), 81-112.
- Coase, R. H. (1937). The nature of the firm. *Economica*, 4(16), 386-405.
- Davidson, R., & MacKinnon, J. G. (1993). Estimation and inference in econometrics. Vol. 63, No. 5 (Sep., 1995).

- De Andres, P., & Vallelado, E. (2008). Corporate governance in banking: The role of the board of directors. *Journal of Banking & Finance*, *32*(12), 2570-2580.
- DeFond, M. L., Hann, R. N., & Hu, X. (2005). Does the market value financial expertise on audit committees of boards of directors? *Journal of Accounting Research*, 43(2), 153-193.
- Demsetz, H., & Lehn, K. (1985). The structure of corporate ownership: Causes and consequences.
- Journal of political economy, 93(6), 1155-1177.
- Eng, L. L., & Mak, Y. T. (2003). Corporate governance and voluntary disclosure. *Journal of Accounting and Public Policy*, 22(4), 325-345.
- Faccio, M., Lang, L. H., & Young, L. (2001). Dividends and expropriation. *American Economic Review*, 54-78.
- Fama, E. F. (1980). Agency Problems and the Theory of the Firm. *Journal* of political economy, 88(2), 288-307.
- Fama, E. F., & Jensen, M. C. (1983). Agency problems and residual claims. *The Journal of Law and Economics*, 26(2), 327-349.
- Fama, E. F., & Jensen, M. C. (1985). Organizational forms and investment decisions. *Journal of Financial Economics*, 14(1), 101-119.
- Felo, A. J., Krishnamurthy, S., & Solieri, S. A. (2003). Audit committee characteristics and the perceived quality of financial reporting: an empirical analysis. Available at SSRN: https://ssrn.com/abstract=401240 or http://dx.doi.org/10.2139/ssrn.401240.
- Filatotchev, I., Lien, Y.-C., & Piesse, J. (2005). Corporate governance and performance in publicly listed, family-controlled firms: Evidence from Taiwan. *Asia Pacific Journal of Management*, 22(3), 257-283.
- Filmer, D., & Pritchett, L. H. (2001). Estimating wealth effects without expenditure data—or tears: an application to educational enrollments in states of India. *Demography*, *38*(1), 115-132.
- Gallery, G., Gallery, N., & Supranowicz, M. (2008). Cash-based related party transactions in new economy firms. *Accounting Research Journal*, 21(2), 147-166.
- Gao, L., & Kling, G. (2008). Corporate governance and tunnelling: Empirical evidence from China. *Pacific-Basin Finance Journal*, 16(5), 591-605.
- Ge, W., Drury, D. H., Fortin, S., Liu, F., & Tsang, D. (2010). Value relevance of disclosed related party transactions. *Advances in Accounting*, 26(1), 134-141.
- Ghani, W., Haroon, O., & Ashraf, M. J. (2010). Business Groups' Financial Performance: Evidence from Pakistan. *Global Journal of Business Research, Vol. 5, No. 2, pp. 27-39. Available at SSRN: https://ssrn.com/abstract=1873553.*

- Gohar, R., & Karacaer, S. (2009). Pakistani business groups: A comparison of group affiliated and unaffiliated firm performance. NUST Journal of Business and Economics, 2(2), 41-53.
- Gomez-Mejia, L. R., Larraza-Kintana, M., & Makri, M. (2003). The determinants of executive compensation in family-controlled public corporations. *Academy of Management Journal*, *46*(2), 226-237.
- Gopalan, R., & Jayaraman, S. (2012). Private control benefits and earnings management: evidence from insider controlled firms. *Journal of Accounting Research*, 50(1), 117-157.
- Gordon, E. A., & Henry, E. (2005). Related party transactions and earnings management. *Available at SSRN: https://ssrn.com/abstract=612234 or http://dx.doi.org/10.2139/ssrn.612234*.
- Gordon, E. A., Henry, E., & Palia, D. (2004). Related party transactions and corporate governance *Corporate Governance* (pp. 1-27): Emerald Group Publishing Limited.
- Gujarati, D., & Porter, D. (2003). Multicollinearity: What happens if the regressors are correlated. *Basic econometrics*, 363.
- Gujarati, D. N., & Porter, D. C. (2009). Causality in economics: The Granger causality test. *Basic Econometrics (Fifth international ed.). New York: McGraw-Hill*, 652.
- Gul, F. A., Kim, J.-B., & Qiu, A. A. (2010). Ownership concentration, foreign shareholding, audit quality, and stock price synchronicity: Evidence from China. *Journal of Financial Economics*, 95(3), 425-442.
- Gulzar, M. A., & Wang, Z. (2010). Corporate governance and non-listed family owned businesses: an evidence from Pakistan. *International Journal of Innovation, Management and Technology, 1*(2), 124.
- Harttgen, K., & Klasen, S. (2012). A household-based human development index. *World Development*, 40(5), 878-899.
- Hill, C. W., & Snell, S. A. (1988). External control, corporate strategy, and firm performance in research-intensive industries. *Strategic Management Journal*, 9(6), 577-590.
- Hill, C. W., & Snell, S. A. (1989). Effects of ownership structure and control on corporate productivity. *Academy of Management Journal*, 32(1), 25-46.
- Hussain, E., & Shah, A. (2015). Impact of ownership structure on dividend smoothing: a comparison of family and non-family firms in Pakistan. *Afro-Asian Journal of Finance and Accounting*, *5*(4), 356-377.
- Ibrahim, A. A. (2006). Corporate governance in Pakistan: Analysis of current challenges and recommendations for future reforms. *Wash. U. Global Stud. L. Rev.*, *5*, 323.

- Ikram, A., & Naqvi, S. A. A. (2005). Family Business Groups and Tunneling Framework: Application and Evidence from Pakistan: Lahore University of Management Sciences. CMER Working Paper Series No. 05-41.
- Irshad, R., Hashmi, S. H., Kausar, S., & Nazir, M. I. (2015). Board Effectiveness, Ownership Structure and Corporate Performance: Evidence from Pakistan. *Journal of Business Studies Quarterly*, 7(2), 46.
- Javaid, F., & Saboor, A. (2015). Impact of Corporate Governance index on Firm Performance: evidence from Pakistani manufacturing sector. *Journal of Public Administration and Governance*, 5(2), 1-21.
- Javid, A. Y., & Iqbal, R. (2008). Ownership concentration, corporate governance and firm performance: Evidence from Pakistan. *The Pakistan Development Review*, 643-659.
- Javid, A. Y., & Iqbal, R. (2010a). Corporate governance in Pakistan: Corporate valuation, ownership and financing. *Working Papers & Research Reports, 2010.*
- Javid, A. Y., & Iqbal, R. (2010b). Corporate governance in Pakistan: Corporate valuation, ownership and financing. Working Papers & Research Reports. PIDE Working Papers 2010: 57.
- Jensen, M. C., & Meckling, W. H. (1976). Theory of the firm: Managerial behavior, agency costs and

ownership structure. Journal of Financial Economics, 3(4), 305-360.

- Jian, M., & Wong, T. J. (2010). Propping through related party transactions. *Review of Accounting Studies*, 15(1), 70-105.
- Jiang, G., Lee, C. M., & Yue, H. (2010). Tunneling through intercorporate loans: The China experience. *Journal of Financial Economics*, 98(1), 1-20.
- Joh, S. W. (2003). Corporate governance and firm profitability: evidence from Korea before the economic crisis. *Journal of Financial Economics*, 68(2), 287-322.
- Johnson, S., La Porta, R., de Silanes, F. L., & Shleifer, A. (2000). Tunnelling, American Economic Review Papers & Proceedings. *American Economic Review*,90(2), 22-27
- Kajola, S. O. (2008). Corporate governance and firm performance: The case of Nigerian listed firms. *European journal of economics, finance and administrative sciences, 14*(14), 16-28.
- Kang, J.-K., & Shivdasani, A. (1995). Firm performance, corporate governance, and top executive turnover in Japan. *Journal of Financial Economics*, 38(1), 29-58.
- Kennedy, P. (2008). A guide to modern econometrics: Oxford: Blackwell Publishing.

- Khan, A., & Awan, S. (2012). Effect of board composition on firm's performance: A case of Pakistani listed companies. *Interdisciplinary Journal of Contemporary Research in Business*, *3*(10), 853-863.
- Khan, A., & Yusof, Z. (2017). Terrorist economic impact evaluation (TEIE) model: the case of Pakistan. *Quality & Quantity*, *51*(3), 1381-1394.
- Khanna, T., & Palepu, K. (2000). Is group affiliation profitable in emerging markets? An analysis of diversified Indian business groups. *The Journal of Finance*, *55*(2), 867-891.
- Klein, P., Shapiro, D., & Young, J. (2005). Corporate governance, family ownership and firm value: the Canadian evidence. *Corporate Governance: An International Review, 13*(6), 769-784.
- Kohlbeck, M., & Mayhew, B. W. (2010). Valuation of firms that disclose related party transactions.

Journal of Accounting and Public Policy, 29(2), 115-137.

- Krishnan, J. (2005). Audit committee quality and internal control: An empirical analysis. *The Accounting Review*, 80(2), 649-675.
- La Porta, R., Lopez-de-Silanes, F., Shleifer, A., & Vishny, R. (1999). The quality of government. *The Journal of Law, Economics, and Organization, 15*(1), 222-279.
- La Porta, R., Lopez-de-Silanes, F., Shleifer, A., & Vishny, R. (2000). Investor protection and corporate governance. *Journal of Financial Economics*, 58(1), 3-27.
- La Porta, R., Lopez-de-Silanes, F., Shleifer, A., & Vishny, R. W. (2000). Agency problems and dividend policies around the world. *The Journal of Finance*, 55(1), 1-33.
- Lauterbach, B., & Tolkowsky, E. (2007). Market-value-maximizing ownership structure when investor protection is weak *Issues in Corporate Governance and Finance* (pp. 27-47): Emerald Group Publishing Limited.
- Leech, D., & Leahy, J. (1991). Ownership structure, control type classifications and the performance of large British companies. *The Economic Journal*, *101*(409), 1418-1437.
- Lehmann, E., & Weigand, J. (2000). Does the governed corporation perform better? Governance structures and corporate performance in Germany. *Review of Finance*, 4(2), 157-195.
- Lei, A. C. H., & Song, F. (2008). Connected transactions and firm valuations: evidence from China affiliated companies. *Pacific-Basin Finance Journal*.
- Lins, K. V. (2003). Equity ownership and firm value in emerging markets. *Journal of Financial and Quantitative analysis*, *38*(1), 159-184.
- Liu, G., & Sun, J. (2010). Director tenure and independent audit committee effectiveness. *International research journal of finance and economics*, 51.

- Lo, A. W., & Wong, R. M. (2011). An empirical study of voluntary transfer pricing disclosures in China. *Journal of Accounting and Public Policy*, 30(6), 607-628.
- Marrakchi Chtourou, S., Bedard, J., & Courteau, L. (2001). Corporate governance and earnings management.
- Maury, B. (2006). Family ownership and firm performance: Empirical evidence from Western European corporations. *Journal of Corporate Finance*, *12*(2), 321-341.
- McConaughy, D. L., Walker, M. C., Henderson, G. V., & Mishra, C. S. (1998). Founding family-controlled firms: Efficiency and value. *Review of Financial Economics*, 7(1), 1-19.
- McMullen, D. A., Raghunandan, K., & Rama, D. V. (1996). Internal control reports and financial reporting problems. *Accounting Horizons*, *10*(4), 67.
- Mehboob, F., Tahir, S. H., & Hussain, T. (2015). Impact of Family Ownership on Financial Decisions of a Firm: An Analysis of Pharmaceutical and Chemical Sectors in Pakistan. *Euro-Asian Journal* of Economics and Finance, 3(2),103-112.
- Morck, R., Shleifer, A., & Vishny, R. W. (1988). Management ownership and market valuation: An empirical analysis. *Journal of Financial Economics*, 20, 293-315.
- Morris, R., Susilowati, I., & Gray, S. (2012). "The Impact of IFRS Adoption versus Non-Adoption on Corporate Disclosure Levels in the Asian Region". Paper presented at the American Accounting Association Annual Meeting and Conference on Teaching and Learning in Accounting, Washington DC. Retrieved from http://aaahq. org/AM2012/display. cfm.
- Morris, R. D., & Gray, S. J. (2007). Corporate transparency differences in the Asian region: The feasibility of global standardization in practice. Paper presented at the AFAANZ Conference, Gold Coast, July. Retrieved from http://wwwdocs.fce.unsw.edu.au/accounting/news/seminar

s2007_s1/Morris_and_Gray.pdf.

- Mudambi, R., & Nicosia, C. (1998). Ownership structure and firm performance: evidence from the UK financial services industry. *Applied Financial Economics*, 8(2), 175-180.
- Nelson, J., Gallery, G., & Percy, M. (2010). Role of corporate governance in mitigating the selective disclosure of executive stock option information. *Accounting & Finance*, *50*(3), 685-717.
- Nicholls, D., & Ahmed, K. (1995). Disclosure quality in corporate annual reports of non-financial companies in Bangladesh. *Research in Accounting in Emerging Economies*, 3(2), 149-170.

- Porta, R., Lopez-de-Silanes, F., & Shleifer, A. (1999). Corporate ownership around the world. *The Journal of Finance*, 54(2), 471-517.
- Porta, R., Lopez-de-Silanes, F., Shleifer, A., & Vishny, R. (2002). Investor protection and corporate valuation. *The Journal of Finance*, *57*(3), 1147-1170.
- Prowse, S. D. (1992). The structure of corporate ownership in Japan. *The Journal of Finance*, 47(3), 1121-1140.
- Remuneration, S. G. o. D., & Greenbury, S. R. (1995). Directors' remuneration: Gee London. Rickling, M. (2014). Audit committee characteristics and repeatedly meeting-beating analyst forecasts. International Journal of Business, 19(2), 173.
- Riyanto, Y. E., & Toolsema, L. A. (2008). Tunneling and propping: A justification for pyramidal ownership. *Journal of Banking & Finance*, 32(10), 2178-2187.
- Roodman, D. (2006a). *How to do xtabond2: an introduction to 'difference 'and 'system.* Paper presented at the GMM in STATA', Center for Global Development Working Paper No. 103.
- Roodman, D. (2006b). How to do xtabond2: An introduction to difference and system GMM in Stata. Center for Global Development Working Paper No. 103. Available at SSRN: https://ssrn.com/abstract=982943 or http://dx.doi.org/10.2139/ssrn.982943.
- Roodman, D. (2009). A note on the theme of too many instruments. Oxford Bulletin of Economics and statistics, 71(1), 135-158.
- Ryngaert, M., & Thomas, S. (2012). Not all related party transactions (RPTs) are the same: Ex ante versus ex post RPTs. *Journal of Accounting Research*, 50(3), 845-882.
- Sahn, D. E., & Stifel, D. (2003). Exploring alternative measures of welfare in the absence of expenditure data. *Review of income and wealth*, 49(4), 463-489.
- Sahn, D. E., & Stifel, D. C. (2000). Poverty comparisons over time and across countries in Africa. *World Development*, 28(12), 2123-2155.
- Sánchez-Ballesta, J. P., & García-Meca, E. (2007). A meta-analytic vision of the effect of ownership structure on firm performance. *Corporate Governance: An International Review*, 15(5), 879-892.
- Sarbah, A., & Xiao, W. (2015). Good corporate governance structures: A must for family businesses. *Open Journal of Business and Management*, 3(01), 40.
- Sarbanes, P. (2002). *Sarbanes-oxley act of 2002*. Paper presented at the Public Company Accounting Reform and Investor Protection Act. Washington DC: US Congress.
- Shleifer, A., & Vishny, R. W. (1986). Large shareholders and corporate control. *Journal of political economy*, 94(3, Part 1), 461-488.

- Shleifer, A., & Vishny, R. W. (1997). A survey of corporate governance. *The Journal of Finance*, 52(2), 737-783.
- Srinivasan, P. (2013). An Analysis of Related-Party Transactions in India. (September 30, 2013). IIM Bangalore Research Paper No. 402. Available at SSRN: https://ssrn.com/abstract=2352791 or http://dx.doi.org/10.2139/ssrn.2352791.
- Stein, J. C. (1997). Internal capital markets and the competition for corporate resources. *The Journal of Finance*, 52(1), 111-133.
- Tahir, S. H., & Sabir, H. M. (2015). Ownership structures as determinants of financial decisions: Evidence from Pakistani family owned listed firms. *Zbornik radova Međimurskog veleučilišta u Čakovcu, 6*(1), 117-127.
- Thomsen, S., & Pedersen, T. (2000). Ownership structure and economic performance in the largest European companies. *Strategic Management Journal*, 689-705.
- Ullah, H., & Shah, A. (2015). Related Party Transactions and Corporate Governance Mechanisms: Evidence from Firms Listed on the Karachi Stock Exchange. Pakistan Business Review, Vol. 17, No. 3, pp. 663-680, 2015. Available at SSRN: https://ssrn.com/abstract=2695278.
- Vafeas, N. (2003). Length of board tenure and outside director independence. *Journal of Business Finance & Accounting*, 30(7-8), 1043-1064.
- Villalonga, B., & Amit, R. (2006). How do family ownership, control and management affect firm value? *Journal of Financial Economics*, 80(2), 385-417.
- Williams, M. P., & Taylor, D. W. (2014). Measuring and Explaining the Extent of Corporate Propping Transactions: Evidence from China. *Global Review of Accounting and Finance*, 5(1), 76-92.
- Williamson, O. E. (1975). Markets and hierarchies. New York, 26-30.
- Wruck, K. H. (1989). Equity ownership concentration and firm value: Evidence from private equity financings. *Journal of Financial Economics*, 23(1), 3-28.
- Wu, S., & Cui, H. (2002). Consequences of the concentrated ownership structure in Mainland China-evidence of Year 2000. Paper presented at the EFA 2002 Berlin Meetings; EFMA 2002 London Meetings; Sauder School of Business Working Paper.
- Xu, X., & Wang, Y. (1997). Ownership structure, corporate governance, and corporate performance: The case of Chinese stock companies (Vol. 1794): World Bank Publications.
- Xu, X., & Wang, Y. (1999). Ownership structure and corporate governance in Chinese stock companies. *China economic review*, *10*(1), 75-98.
- Zalewska, A. (2014). Challenges of corporate governance: Twenty years after Cadbury, ten years after Sarbanes–Oxley: Elsevier.